Author’s response to reviews

Title: Gender differences among Indigenous Canadians experiencing Homelessness and Mental Illness

Authors:

Brittany Bingham (bld@sfu.ca)
Akm Moniruzzaman (akm_moniruzzaman@sfu.ca)
Michelle Patterson (michelle_patterson@sfu.ca)
Jitinder Sareen (sareen@umanitoba.ca)
Jino Distasio (j.distasio@uwinnipeg.ca)
John O’Neil (joneil@sfu.ca)
Julian Somers (jsomers@sfu.ca)

Version: 1 Date: 01 Nov 2018

Author’s response to reviews:

Thank you for this thoughtful review. The requested reviewer changes to ref no. 65 have been addressed in the manuscript. Respectfully, Brittany Bingham